MOSS 2 Debrief Minutes

August 5, 2003

ELEMENT ATTENDANCE:

ESDIS SSDO, MRTT, EDOS, EMOS, ASDC DAAC, TES SIPS, GES DAAC, MLS SIPS, HIRDLS SIPS, OMI SIPS, Averstar

NEXT TELECON: September 2, 2003 at 2pm EST in Bldg 32 room S241

DIAL-IN NUMBER:

USA Toll Free Number: 888-395-8106

PASSCODE: 92854

MOSS 3 (5 day test) Data Preparation Schedule:
· New data due to ESDIS by August 15th
· Write tapes week of Sept 8th;Verify tapes week of Sept 15th; Ship tapes Sept 22nd
· Dry Run - October 6-8 

· TRR - October 14th  (regularly scheduled meeting)

· Formal Run - October 20-24

· Requirement Status Complete Oct 31st
· Debrief Nov 4th (not regularly scheduled meeting) 

· MOSS telecon Nov 25th cancelled

· SORR Dry Run - Nov 12th? or 18th? (TBD)

· SORR Formal Run - December 2nd
MEETING HIGHLIGHTS:

All debrief presentations can be obtained from the Aura Testing website: http://jupiter02.gsfc.nasa.gov/AuraTesting/ under MOSS2 link. Only highlights, changes, or clarifications will be presented below.

SSDO:

Karen discussed the MOSS 3 schedule presented above. In summary, new data are expected from OMI and TES teams and are due to ESDIS by August 15th; the dry run will be a full dry run (5 days) starting with the ground stations; formal run scheduled for Oct 20-24; requirements status complete by Oct 31; and debrief Nov 4th in order to support the SORR Dry Run. Karen asked all to review the schedule and let her know if there are any problems with the schedule.

Karen mentioned that the requirement charts with the element percentages currently do not reflect the requirements verified in MT #1. They will be updated for the Interim SORR on August 12th.

It was decided to try to schedule play back of the missing MT #1 data this Thursday August 7th. This would mainly involve the GES DAAC and corresponding SIPS. ASDC and TES may see a little data. The re-delivery of the modified p-field data to ASDC for TES is tentatively planned for next Thursday August 14th.

Karen mentioned that the MT #1 Full Science Debrief would be September 9th. All teams indicated that they could support that date (i.e., provide Karen the data to be presented). Karen mentioned that the Aura project would like more information on the science processing: processing successful, anything that prevented processing, any instrument problems found that were not known, and any feedback that they want reported back to the Aura project.

Karen asked EMOS if the MT#1 history data could be sent to the DAAC and Utah responded that it could. They have to configure one more system and then they would be ready. GES DAAC stated that they believe to be ready to ingest in TS2 and will have polling turned on to ingest the products.

For CPT3 data, a date will be arranged after EDOS completes their analysis. There could be some overlap of data dates with the MT#1 data.

MRTT: 

Susan presented the MRTT slides and mentioned that 103 out of the 150 MRTT requirements passed; 4 passed with workarounds; and 43 remains untested. No failures reported. Karen agreed to send Susan EMOS’s MT#1 data listing to possibly assist with MRTT requirements verifications.

FDS: 

Not present. FDS provided the slides prior to the debrief. Their data were pre-delivered to the GES DAAC prior to MOSS2 and were successful processed.

EDOS:

Ron presented the EDOS slides. EDOS passed 62 out of the 63 MOSS 2 requirements with 1 untested. No major problems found. See the presentation for issues/problems encountered during the test. EDOS also verified their MOSS 3 requirements to the HIRDLS Oxford University during MT #1.

EMOS:

Utah presented his slides via telecon. No problems found. The original MOSS 2 requirements were reallocated to Spacecraft tests and passed. Utah mentioned that they got everything that were expected. He stated that 3 carryout files failed and were not sent to GES DAAC due to minimum file size issue.

GES DAAC:

Ed reported that the GES DAAC experienced no major problems during MOSS2.  The GES DAAC reported that early in the test they had difficulties in ingesting HIRDLS and MLS higher-level products and at the time of the meeting, they had not successfully ingested the OMI SIPS higher-level products. The problem turns out to be a GES DAAC internal problem, trouble ticket TT-10047 written, and Landover is involved.  After the telecon: As per Ed McDonald, the TS2 ingest problems were resolved (8/7/03) and they are working with OMI SIPS in resolving metadata issues. They appear to be related to OMI NOSE.

Action Item (AI): Ed and Vince are to check into/verify the delivery of the HIRDLS production history file (PH) to the DAAC.

OMI SIPS:

Elaine presented the slides via telecon. She identified as problems: MTMGW orders and GES DAAC transmission problems for L1B and L2 products. OMI SIPS was able to stage two products (OML1BCAL and OML1BIRR) for ODPS to pick up.

HIRDLS:

Vince presented his slides via telecon.  HIRDLS passed 14 out of their 19 requirements; failed 2 and not test 3. It was recommended that requirement 1325 be status as ‘Untested” due to the software not being ready; could not be tested and #1301 be status as “Passed” since one EDS had been delivered from the DAAC via the interface. Thus leaving 1 failed; 20 passed; and 2 untested (ESDTs not in OPS). Vince also mentioned that HIRDLS expected 43 science granules for HIR0SCIX but received only 42. Bill explained that it was due to the quality flag (QL) not set in SCS #18. Vince will work with Ed to locate their PH in the GES DAAC. After the telecon: ESDIS made the updates to reqs 1325 and 1301.

Karen asked Ed of the GES DAAC when will the 2 ESDTs would be promoted to OPS. Ed replied that the CCR would have to be written and approved by the CCB before they could be promoted. Now that they have test data for the ESDTs the verification process by James Johnson can begin.

MLS SIPS:

David presented the slides via telecon. He summarized the issues that prevented MLS from testing more of their MOSS 2 requirements. He stated that they did not have sufficient time to integrate and test the Toolkit patch 5.2.9.1.b; and their L1 did not produce meaningful output for L2 processing. David mentioned that they reported a new Toolkit problem to ECS today and that all other problems have been fixed. Few MLS internal problem reports were also written. Many have been corrected.

ASDC DAAC:

Steve presented the slides via telecon. He mentioned that ASDC passed 78 of 84 requirements; 3 failed and 3 untested. The failures were due to OPS re-ingest GUI tool need to be updated for expedited data. Tracking Ticket 6212 was written and will be fixed before MOSS 3. The untested requirements could not be verified due to the ingest failures of the APIDs and therefore data could not be distributed to the SIPS. The previously failed MOSS 1 requirements were also verified and passed.

TES SIPS:

Andy presented the slides via telecon and reported that 32 out of the 35 MOSS 2 requirements passed and 3 were untested.  The untested requirements were due to not receiving EDSs for APIDs 1669, 1670, and 1671. Andy also mentioned that he will double check his counts of products sent to the ASDC DAAC because there were some discrepancies in the two sets numbers.

Averstar:

Scott presented the slides via telecon.  (AI): Scott mentioned that he could not find the HIRDLS product PH (production history) but agreed to perform a different type of search to locate the product. According to HIRDLS, the product is at the GES DAAC. It was mentioned that the EMOS history product AUMMSARC plans to be deleted in the near future.
ACTION ITEMS: Documented in the Aura Action Item Tool at http://jupiter02.gsfc.nasa.gov/auratesting/ .  Please remember to enter your status in the tool.

PAGE  
4

